Author's response to reviews

Title: Completeness and timeliness of Salmonella notifications in Ireland in 2008: a cross sectional study

Authors:

Nathalie Nicolay (nathalienicolay@yahoo.fr)
Patricia Garvey (patricia.garvey@hse.ie)
Niall DeLappe (niall.delappe@hse.ie)
Martin Cormican (martin.cormican@hse.ie)
Paul McKeown (pauljmckeown@hse.ie)

Version: 4 Date: 13 August 2010

Author's response to reviews: see over
12 August 2010

The Editor
Biomed Central

Re: 6314984003631631 - Completeness and timeliness of Salmonella notifications in Ireland in 2008: a cross sectional study

Dear Sir/Madam
This is to confirm that my colleagues and I have permission to use the data in the study which is non patient specific notification data and is openly available for analysis to designated public health staff involved in investigation and control of infectious diseases. We do not need to make any changes to manuscript and therefore I have not uploaded a new version.

Yours sincerely

Martin Cormican